Author's response to reviews

Title: Associations between socioeconomic factors and fruit and vegetable consumption of children in grades five and six in British Columbia, Canada.

Authors:

Adrienne Attorp (aattorp@gmail.com)
Jenny Scott (jescott@sfu.ca)
Ann C Yew (ann_yew@sfu.ca)
Ryan Rhodes (rhodes@uvic.ca)
Susan I Barr (susan.barr@ubc.ca)
Patti-Jean Naylor (pjnaylor@uvic.ca)

Version: 2 Date: 15 March 2013

Author’s response to reviews:

I am finalizing my submission but the note sent me back to reload our cover letter. Apologies.